Author's response to reviews

Title: Assessing implementation fidelity of a community-based infant and young child feeding intervention in Ethiopia identifies delivery challenges that limit reach to communities: a mix-method process evaluation study

Authors:

Sunny S Kim (sunny.kim@cgiar.org)
Disha Ali (disha_nan@yahoo.com)
Andrew Kennedy (a.l.kennedy@cgiar.org)
Roman T Gebremedhin (r.tesfaye@cgiar.org)
Amare W Tadesse (amareworku@addiscontinental.edu.et)
Teweldebrhan H Abrha (teweldebrhan.hailu@gmail.com)
Rahul Rawat (r.rawat@cgiar.org)
Purnima Menon (p.menon@cgiar.org)

Version: 2
Date: 15 October 2014

Author’s response to reviews:

As per instructions, line numbering was added, and a statement of written informed consent was included in the Methods.